Author’s response to reviews

Title: Tuberculosis burden in China: A high prevalence of pulmonary tuberculosis in household contacts with and without symptoms

Authors:

Zhongwei jia Dr. (urchinji@163.com)
shiming cheng professor (smcheng@chinatb.org)
Tianhao zhang (tianhao14032@163.com)
yan ma (mayan0825@sina.com)
Liqiong bai (hualin_yang@163.com)
Weiguo Xu (weiquoxu_xu@163.com)
Xiaoxin He (hexiaoxinbj@126.com)
Jinkou Zhao (jinkou.zhao@theglobalfund.org)
David Christiani (dchris@hsph.harvard.edu)
Peiru Zhang (liugang_liu2011@163.com)

Version: 4 Date: 19 November 2013

Author’s response to reviews:

Dear Dr. Mayanja:

We are grateful for all the work done by the editors and reviewers on our paper “Tuberculosis burden in China: a cross-sectional analysis in the household contacts of tuberculosis patients”, which has led to a better article. We replied to all comments point by point. The comments made to us are in plain text, and our replies are in bold. The principal changes to the main text are highlighted in yellow on the marked version. We remain open, of course, to further suggestions for improvement.

Many thanks
Zhongwei